

# The determinants of the financial distress of Italian municipalities: How much is it due to inadequate resources?

Emanuele Padovani<sup>1</sup> · Francesco Porcelli<sup>2</sup> · Alberto Zanardi<sup>3</sup>

Accepted: 28 June 2024 / Published online: 24 July 2024 © The Author(s) 2024

## Abstract

The key question addressed in this work is whether financial distress recently experienced by several Italian municipalities can be at least partially imputed to the inadequacy of the financial resources they suffer compared to the needs of their populations and their territories. Starting from a multidimensional definition of financial distress, we investigate this issue by exploiting the variability across municipalities revealed by two different occurrences recently involving the intergovernmental fiscal relations in Italy: on the one hand, the large cuts in vertical transfers carried out by the central government as part of the fiscal consolidation strategy in the period 2014–2015 which affected single municipalities to varying degrees; and, on the other hand, the introduction of a new mechanism of equalization transfers at municipal level which showed how the gap between available financial resources to local needs is differentiated across municipalities. Exploiting these sources of variability across local authorities, the estimation results show that the Italian municipalities which suffer a level of resources lower than that necessary to provide public services at a standard level are, ceteris paribus, more likely to run into financial difficulties. By the same token, large cuts in central government transfers have a statistically significant effect on financial vulnerability at municipal level.

Keywords Financial distress · Resources inadequacy · Local government · Italy

Alberto Zanardi alberto.zanardi@unibo.it

> Emanuele Padovani emanuele.padovani@unibo.it

> Francesco Porcelli francesco.porcelli1@uniba.it

<sup>1</sup> Department of Management, University of Bologna, Bologna, Italy

<sup>2</sup> Department of Law, University of Bari, Bari, Italy

<sup>3</sup> Department of Economics, University of Bologna, Bologna, Italy

#### JEL Classification H70 · H72 · H77

## 1 Introduction

Due to the deterioration of their public finance position, during and after the 2008 Great Recession many local governments experienced severe financial difficulties and sometimes were compelled to declare financial insolvency. The US experience offers several outstanding cases of large municipalities which filed for bankruptcy, such as Stockton and San Bernardino in 2012 and Detroit in 2013 (Beckett-Camarata & Grizzle, 2014; Gorina et al., 2017). Even In Europe, where bailout procedures are much more common (Allers & de Natris, 2021), cases of financial distress of subnational authorities have been frequent. The most relevant episode was that of the Catalonia government in 2012, whose financial crisis can be traced back to a combination of cuts in central transfers, local economy recession and effects of real estate speculation (De La Fuente, 2022; Díaz Mendoza et al., 2015), but other significant cases have occurred elsewhere, such as in Portugal (Lobo et al., 2011).

Italy, which together with Austria, Hungary and Switzerland, applies a stricter insolvency regime (Person, 2021), was no exception. It was severely hit by the 2008 Great recession and local governments, which are responsible for about one third of total public spending, had to deal with harsh direct and indirect consequences (Ambrosanio et al., 2016). Cuts in central transfers to subnational governments have been accompanied by a strengthening of local fiscal autonomy and a tightening of rules on budget discipline. All these financial tensions brought to a drop in local investments and to a rapid increase in the number of municipalities in budget difficulties. If in the 12-year period before the Italian sovereign debt crisis between 1998 and 2010, 71 local governments experienced fiscal distress, in the following 12 years (2011–2022) as many as 291 local governments were involved in one procedure of legal default.<sup>1</sup> All in all, local governments located in Southern Italy were more affected by financial difficulties (Banca d'Italia, 2022).

Starting from this evidence, the aim of this paper is to empirically investigate whether the cases of financial distress recently experienced by several Italian municipalities can be at least partially imputed to the inadequacy of the financial resources they suffer compared to the needs of their populations and their territories. To explore this issue, we exploit the variability across municipalities revealed by two recent occurrences involving the intergovernmental fiscal relations in Italy. On the one hand, we make reference to the large cuts in vertical transfers carried out by the central government as part of the fiscal consolidation strategy in the period 2014–2015 which affected the revenues of single municipalities to varying degrees. On the other hand, the introduction in 2015 of a new mechanism of equalization transfers at local level, based on municipal expenditures needs indicators estimated on the basis of structural characteristics of each local entity, has made it possible to get a measure, exogenous from the effects on the budget of local policy decisions,

<sup>&</sup>lt;sup>1</sup> Among the main municipalities, Naples entered into legal financial distress in 2012, while Rome established a sort of "bad company" to manage bad debt as of April 2008.

of the gap between available financial resources to local needs in each municipality. To corroborate our exogeneity assumption, it should be noted that municipalities did not receive any information on the structural equalization gap and could not anticipate its impact on their budget policy. Moreover, the full implementation of the new equalization system involves a very long transitionary period, up to 2030.

By exploiting these two sources of variability across local authorities, the estimation results show first of all that the Italian municipalities which suffer a level of resources lower than what would be necessary to provide public services at a standard level, as measured by the estimated expenditures needs indicators, are, ceteris paribus, more likely to run into financial difficulties. By the same token, large cuts in central government transfers have a statistically significant effect on financial vulnerability at municipal level.

The literature dealing with the determinants of financial distress of public institutions at the subnational level is quite extensive (see Sect. 2). This paper contributes to this literature in two ways. On the one hand, it explores the role specifically played by the inadequacy of financial resources compared to public expenditure needs in triggering cases of financial crisis for some municipalities, a dimension that has not yet been explored in depth by the existing literature. On the other hand, this issue is addressed on the empirical ground following a novel approach which exploits an exogenous measure of the resources gap suffered by individual municipalities provided by a recent institutional innovation in the Italian context.

The structure of the paper is as follows. Section 2 provides a review of the literature about the determinants of financial distress in subnational governments and how our paper is placed within this framework. In Sect. 3 we discuss the legal cases of financial crisis at local level in the Italian institutional setting and the broader, multidimensional definition of financial distress we adopt in this analysis. Section 4 deals with the two measures of inadequacy of financial resources we define starting from two relevant occurrences in the recent developments of the intergovernmental financial relations in Italy that is, as mentioned before, the cuts in transfers for central government to municipal level and the estimation of expenditures needs indicators as a part of the newly introduced mechanism of municipal equalization as a standardized measure of financial needs of each local authority. In Sects. 5 we present the data. Section 6 describes the empirical model to estimate the role of inadequacy of financial resources in affecting the probability of financial distress. Section 7 presents the main empirical results and some relevant robustness checks. Section 8 offers some concluding remarks.

## 2 Factors causing financial distress in municipalities

#### 2.1 Financial distress

Financial distress represents the negative side of financial health in municipalities, opposite to financial sustainability, which implies a stable, positive financial condition (Gardini & Grossi, 2018). Financial distress occurs when a government

struggles to meet its financial obligations and is often referred to as fiscal health, financial position, fiscal stress, or fiscal crisis (Cabaleiro et al., 2012). Definitions vary: some describe it simply as the inability to meet obligations, while others include criteria like acceptable taxation levels, service provision, and both short-term and long-term financial perspectives (Carmeli & Cohen, 2001; Kloha et al., 2005; Wang et al., 2007; Jones & Walker, 2007; Zafra-Gomez et al., 2013).

In this study, financial distress is defined as the inability to balance the budget and provide services and meet future obligations (Pagano & Moore, 1985). Four dimensions of municipal financial distress are widely acknowledged (Groves et al., 2003; Hendrik, 2011; Jacob & Hendrick, 2012): (1) Cash solvency: Assesses liquidity and cash management, focusing on meeting immediate liabilities like payroll and supplier payments. Poor cash management leads to payment delays and service disruptions. (2) Budget solvency: Determines if annual revenues cover expenditures without resorting to borrowing or reserves. Deficits indicate financial distress. (3) Long-run Solvency: Considers long-term obligations' impact on future resources, assessing financial sustainability. Absence of long-term solvency burdens future generations. (4) Service-level solvency: Measures the ability to deliver required public services, impacting service quality and quantity. Cutting services to improve finances risks dissatisfaction and falling below service standards.

#### 2.2 Structural, socio-economic factors

Literature has focused on the conditions of financial distress in municipalities, categorized into structural (external) and non-structural (internal) factors (Gardini & Grossi, 2018; Pammer, 1990). Structural factors are exogenous constraints beyond local politicians' control but significantly impact financial distress. Non-structural factors are within the control of local policymakers and reflect their choices. Structural factors include socio-economic and institutional design elements.

One structural factor is declining population: as the population decreases, so does the tax base, leading to a mismatch between service expenses and revenue, exacerbated by "cost-stickiness", where service expenditures remain unchanged despite population decrease, leading to financial distress (Cabaleiro et al., 2012; Capalbo & Grossi, 2014; Cohen, 2008, 2017; Rodriguez-Bolivar et al., 2016).

Another structural factor is local economic shocks, which negatively impact the tax base, reduce government revenue, and increase welfare program demands (Garcia-Sanchez et al., 2012, 2014; Ashworth et al., 2005; Capalbo & Grossi, 2014; Rodriguez-Bolivar et al., 2016). During crises like the pandemic, higher financial vulnerability in local governments can be mitigated by central state emergency policies (Padovani et al., 2021, 2022).

#### 2.3 Structural, institutional design factors

Literature identifies four institutional design factors affecting financial distress: fiscal rules, central government mandates, local tax autonomy, and inadequacy in financial resources. Fiscal rules, including bail-in versus bail-out procedures, debt ceilings, and spending review policies, influence local financial stability. However, their impact is unclear due to opportunistic behavior by policymakers expecting central government support in financial crises (soft budget constraint) (Raffer & Ponce, 2021; Ter-Minassian, 2007).

Central government mandates can also significantly impact local financial stability by making it difficult to reduce expenditures when revenues decrease, leading to imbalances (Ahrens & Ferry, 2020; Nemec & Spacek, 2020).

Financial autonomy, or the decentralization of income and expenditure decisions, can enhance local government efficiency and effectiveness (Ladner et al., 2016; Oates, 1972). While debated, there is consensus that increased financial autonomy reduces financial vulnerability (Navarro-Galera et al., 2017; Ahrens & Ferry, 2020).

Finally, inadequacy in financial resources can contribute to financial distress in local governments. This inadequacy arises when the revenue capacity of local governments does not match their expenditure needs, making it difficult to generate sufficient revenue to meet financial obligations and fund essential public services. Kloha et al. (2005) highlight how revenue constraints can limit the ability of local governments to effectively manage their budgets. Similarly, Pagano and Moore (1985) discuss the challenges faced by local governments when their fiscal capacity is insufficient to address the demands for public services and infrastructure. Inman (1995) further elaborates on the structural financial imbalances that can lead to persistent fiscal stress, emphasizing the need for adequate fiscal policies and intergovernmental transfers to bridge the gap between revenue and expenditure requirements.

#### 2.4 Non-structural, non-financial factors

Non-structural factors can be categorized as non-financial and financial. Nonfinancial factors include political aspects, mismanagement, and outsourcing and decentralization strategies. Political factors impacting financial distress include the local government's ideological leaning (Brusca et al., 2015; Garcia-Sanchez et al., 2014; McDonald III, 2015) and the strength of the political majority (Ashworth et al., 2005; Solé-Ollé, 2006). These factors influence policies and financial health. Election timing can disrupt financial planning (Ashworth et al., 2005; Garcia-Sanchez et al., 2014), and alignment with the central government affects support and funding (Ahrend et al., 2013).

Mismanagement, such as long-term personnel stability or inadequate training, leads to inefficiencies, higher expenses, lower revenues, increased debt, and tax collection issues (Carmeli & Cohen, 2001; Wallsted et al., 2014).

Outsourcing and decentralization strategies can both improve efficiency and reduce costs but also reduce control and accountability, complicating coordination across government levels (Cuadrado-Ballesteros et al., 2013; Zafra-Gomez et al., 2013).

Structural Socioeconomic Declining populati Shocks on local ec Institutional Design of fiscal ru	h	
socioeconomic Deciming populati Shocks on local ec Institutional Design of fiscal ru	omy	
Shocks on local ec Institutional Design of fiscal ru	omy	ohen (2008, 2017), Cabaleiro et al., (2012), Capalbo and Grossi (2014) and Rodriguez-Bolivar et al. (2016)
Institutional Design of fiscal ru		vshworth et al. (2005), Garcia-Sanchez et al., (2012, 2014), Capalbo and Grossi (2014), Rodriguez- Bolivar et al. (2016) and Padovani et al., (2021, 2022)
		affer and Ponce (2021) and Ter-Minassian (2007).
Central governmer	nandates	whrens and Ferry (2020) and Nemec and Spacek (2020)
Financial autonom		Jhapman (1988), Ladner et al. (2016), Oates (1972), Navarro-Galera et al. (2017), Ahrens and Ferry (2020) and Mikesell (2013)
Inadequacy of fin	cial resources	Cloha et al. (2005), Pagano and Moore (1985) and Inman (1995)
Non-structural		
Non-financial Political factors		vshworth et al. (2005), Solé-Ollé (2006), Garcia-Sanchez et al. (2014), Ahrend et al. (2013), Brusca et al. (2015) and McDonald III (2015)
Mismanagement		armeli and Cohen (2001) and Wallsted et al. (2014)
Outsourcing and d	entralization	afra-Gomez et al. (2013), Cuadrado-Ballesteros et al. (2013)
Financial Revenue autonomy	nd diversification	ones and Walker (2007), Martell (2008), Trussel and Patrick (2009), Cabaleiro et al. (2012) and Lara-Rubio et al. (2017)
Expenditure rigidi	S	acob and Hendrick (2012) and Cohen et al. (2017)
Opacity of account	g information	enito and Bastida (2004), Rodriguez-Bolivar et al. (2014), Padovani and Rescigno (2018)
Quality of budgeta	process	armeli (2008), Kimhi (2008) and Tang et al. (2014).

 Table 1
 Factors causing financial distress in local governments



Fig. 1 Municipalities involved in legal insolvency procedures (ordinary regions, cumulative 2014-2023)

#### 2.5 Non-structural, financial factors

Financial non-structural factors include low revenue autonomy and diversification, expenditure rigidities, opacity of accounting information, and low-quality budgetary processes. Low revenue autonomy means local governments rely on limited revenue sources, making them vulnerable to economic shocks and changes (Mikesell, 2013). Limited revenue-generating capacity negatively impacts financial health and sustainability (Jones & Walker, 2007; Martell, 2008; Trussel & Patrick, 2009; Lara-Rubio et al., 2017). Chapman (1988) noted that local governments face more risk when funding decisions are made by others, leading to financial vulnerability (Cabaleiro et al., 2012).

Expenditure rigidities, like personnel expenses and debt service payments, hinder local governments' ability to adjust expenditures during revenue reductions, causing imbalances and financial distress (Cohen et al., 2017; Jacob & Hendrick, 2012). As discussed in previous literature (Padovani et al., 2021), this factor and the low revenue autonomy are interconnected with institutional factors. Specifically, decisions regarding revenue diversification and expenditure rigidities at the municipal level are often influenced by the country's fiscal rules, central government mandates, and the level of financial autonomy. In essence, while decisions are made locally, they are constrained by national-level boundaries and regulations.

Opacity in accounting information prevents stakeholders from accurately assessing financial health, complicating efforts to address financial issues (Benito & Bastida, 2004; Rodriguez-Bolivar et al., 2014; Padovani et al., 2018).

A low-quality budgetary process, including poor planning and resource use, contributes to financial distress by weakening financial management (Carmeli, 2008; Kimhi, 2008; Tang et al., 2014). A synopsis of the main factors causing financial distress is portrayed in Table 1.

Levine et al. (2013) and Maher et al. (2023) extensively explore factors contributing to financial distress in US cities. In the European context, Spain has received significant attention, with studies by Cohen (2008), Cabaleiro and Buch (2011), Cuadrado-Ballestreros et al. (2013), and Brusca et al. (2015). Fewer studies focus on Italian municipalities (Brusca et al., 2015; Bisogno et al., 2019; Capalbo & Grossi, 2014; Cuadrado-Ballesteros et al., 2013; Gregori & Marattin, 2019). The imbalance between revenue capacity and spending needs, termed inadequacy of financial

resources, is identified as a key factor by Kloha et al. (2005), Pagano and Moore (1985), and Inman (1995). However, empirical tests to measure its effects are lacking, presenting a research gap we aim to fill in this study.

## 3 Financial distress in Italian municipalities: institutional setting

The Italian the legal framework provides for three different cases of financial distress and of the related financial recovery procedures for local authorities, which impose an increasingly greater restriction on their autonomy: structural deficit, intermediate pre-default and financial default.

Local authorities that are structurally in deficit—a condition that is established on the basis of a series of predetermined parameters—are subjected to central government controls relating to staff and the costs of certain services. Local authorities incurring a situation of structural deficit, that can evolve into financial default, can autonomously decide to launch the intermediate pre-default procedure. This procedure leaves financial management in the hands of the elective local policymakers instead of transferring all powers to a commissioner, even if the local body is subjected to intensive controls aimed at preventing a possible future bankruptcy. Lastly, financial default occurs if a local authority is no longer able to carry out its functions and to deliver essential services, or is unable to meet its financial obligations. The recovery process is managed both by the local government, as regards ordinary management, and by a special liquidation authority, which deals with debt management.<sup>2</sup>

Our analysis of the determinants of financial distress in the Italian municipalities could, as a consequence, be based directly on cases of legal default and of the other instances of financial crisis that can be derived from public records. However, referring to these legal insolvency regimes suffers from a number of drawbacks. First, in the recent experience of the Italian municipalities, a strong propensity of local authorities to postpone as much as possible the official declaration of financial distress clearly emerges (Ambrosanio et al., 2016; Person, 2021). Second, several municipalities avoided falling into insolvency thanks to massive cash advances provided by banks (Raffer & Padovani, 2019). Finally, Italian municipalities show a

<sup>&</sup>lt;sup>2</sup> A reform of the different procedures of financial distress above illustrated is currently under discussion in Italy. The main elements of the proposed reform include unifying the multi-year financial rebalancing procedure and the default one, excluding municipalities in financial difficulty from the fiscal consolidation measures implemented by the central government, and assigning financial support to those local entities which incur financial distress outside the direct responsibility of local policy makers.

long record of bail-out carried out by the central government through huge transfers to avoid bankruptcy (Raffer & Padovani, 2019).

As a result, only the tip of the iceberg of the financial distress experienced by the Italian municipalities is actually captured by the legal cases: Fig. 1 shows that in the entire period from 2014 to 2023 only 3% and 5% of municipalities located in ordinary regions were involved in legal procedures of financial default and intermediate pre-default respectively (244 and 401 municipalities in total). Moreover, most of these cases occurred in municipalities located in Southern Italy.<sup>3</sup>

We sought to overcome these shortcomings by relying on a more multidimensional measure of financial distress at the municipal level. In the case of the Italian municipalities, AIDA-PA database provides a financial distress scoring system for individual institutional bodies, named *Rating Finanziario*, which can prove useful in this perspective.<sup>4</sup> In particular, a set of ten elementary financial indicators/ratios are derived from the budgetary modified-accrual accounting reports (*rendiconti finanziari*) of each municipality. These elementary financial ratios are able to capture the three most relevant dimensions of financial conditions at municipal level mentioned in Sect. 2: cash solvency, budget solvency and long-run solvency.<sup>5</sup> Moreover, a Financial Rating Index is derived as a summary indicator of overall financial health by combining the elementary financial ratios (Financial Rating Index is normalized in the range 0–10).

Within the array of the indexes set out by AIDA-PA database, we selected, in addition to the overall Financial Rating Index, two elementary indexes which we consider particularly relevant to catch the different dimensions of financial health or, on the contrary, of the financial distress at local level:

- the Structural Current Equilibrium Index (R4) is defined as the ratio of current revenues (deducted of bad debt) to current expenditures plus loans repayment, and provides a summary measure of budget solvency. This measure captures the capacity of a municipality to cover ordinary expenditures, namely those for goods and services for current activities including the debt service, with similar revenues, that is taxes, fees and current grants;
- 2. the Use of Cash Advances index (R8) is defined as the ratio of cash advances to current revenues, and provides a summary measure of the cash solvency. Cash advances are short term cash facilities used by municipalities in case they run out of cash. A high level, or a frequent use, of cash advances is symptomatic of ineffective cash management, usually caused by incapacity to collect own revenues (Corte dei conti, 2022).

<sup>&</sup>lt;sup>3</sup> Table 9 in the "Appendix" reports all official cases of intermediate pre-default and financial default from 1989 to 2023.

<sup>&</sup>lt;sup>4</sup> This scoring system is used by the Italian Court of auditors (Corte dei conti, 2021) and several public and private institutions to assess financial risk of municipalities.

<sup>&</sup>lt;sup>5</sup> Service solvency is not measured since standard levels of public services provision at municipal level have not yet been established.

	Population	Transfer Cuts Effect (TCE) (euro, per capita)	Standard Expenditure Needs (a) (euro, per capita)	Financial resources 2013 (b) (euro, per capita)	Equalization Gap Effect (EGE) (a)–(b) (euro, per capita)
Giuliano in Campania	124,361	23	300	210	89
Perugia	165,683	32	321	319	1
Salerno	133,970	50	290	462	- 172

Table 2 Transfer cuts effect and equalization gap effect in an illustrative sample of municipalities

We did not select any specific index to capture long-run solvency since this dimension seems not to be relevant for the Italian sub-national governments for three different reasons. First, the public debt of local governments in 2023 accounts for only 1.27% of the total public debt and this share is constantly decreasing over time (from 2017 and 2022 the consolidated long-term debt in municipalities fell by 18 p.p.). Moreover, cash advances have sometimes been used by local authorities as improper long-term debt instruments, especially in municipalities in Southern Italy. Lastly, the index R4 inherently considers in the denominator the debt burden, which include loans repayment. Therefore, the two selected indexes already absorb the financial distress dimension related to long-run solvency.

## 4 The measurement of financial resources inadequacy and performance

As mentioned above, the focus of this paper is on the role of the inadequacy in financial resources in determining its financial distress of single municipalities.

In the recent evolution of public finances in Italy the issue of the inadequacy of financial resources of local governments compared to what would be necessary for the fulfillment of their public functions has been raised in connection with at least two important occurrences that involved the municipal level of government.

The first concerns the large cuts in vertical transfers to municipalities carried out by the central government as part of the fiscal consolidation strategy in the period 2014–2015. These cuts (about 2 billion euros in total) were burdened on all municipalities roughly in proportion of their receipts (with the exemption of municipalities hit by the 2009 and the 2012 earthquakes). Given that the funding of municipal expenditures largely still depends on intergovernmental grants (vertical fiscal imbalance) in addition to decentralized taxes, the decision of the central government to shift part of the fiscal adjustments to local governments through cuts in intergovernmental transfers may have pushed municipalities to run into financial difficulties.

Therefore, in the empirical exercise developed below we defined a Transfer Cuts Effect (TCE) measured for each municipality as the total cut in

Table 3         Expenditure gap, output           gap and efficiency indices         in an illustrative sample of           municipalities         in an illustrative sample of		Population	Expenditure gap (EG) (%)	Output gap (OG) (%)	Efficiency index (EI) (%)
	Giuliano in Campania	124,361	17	-51	- 17
	Perugia	165,683	8	27	18
	Salerno	133,970	-12	-25	- 18

intergovernmental transfers recorded in the period 2014–2015 and we assessed whether it could be significant in determining the financial distress of municipalities.

The second occurrence relevant for the issue of assessing the inadequacy in financial resources at municipal level was the introduction in 2015 of a new mechanism of equalization transfers for Italian municipalities, the so-called Municipal Solidarity Fund-MSF (Arachi et al., 2023). This reform provides for a progressive implementation of a system of equalization of expenditure needs which aims to fill the gap in each municipality between its estimated expenditure needs and its standard tax capacities. In particular, the estimate of Standard Expenditure Needs indicators (SENs) for the main local expenditure programs provides a measure for each municipality of the standard resources deemed necessary to carry out main municipal public functions on the basis of the structural socio-economic conditions affecting production costs and the demand for local public services, regardless of the discretionary choices of local policymakers. Here we take advantage of the implementation of the new equalization mechanism to derive a measure of the inadequacy/excess in the financial resources possibly suffered/benefited by each municipality by comparing the resources actually available in 2013 (the baseline year for the implementation of the equalization mechanism, that since 2013 is kept constant) to what it should have in the case of full achievement of SENs, which will actually occur in 2030 at the end of a long transition period. Given that the evaluation of the gap in the availability of financial resources is derived with reference to a level-the full financing of estimated expenditure needs in 2030 as calculated in 2013—which is not influenced by the policy choices actually implemented by local policy-makers in the period we analyze (2016-2019), the strategy applied here can provide an exogenous measure of the possible inadequacy in the financial resources at municipal level. To corroborate our exogeneity assumption, it should be noted that municipalities did not receive any information on the structural equalization gap and could not anticipate its impact on their budget policy.

In particular, in the empirical exercise we defined an Equalization Gap Effect (EGE) measured for each municipality as the difference between the full financing of SENs and the resources available in 2013 and we assessed whether it could be significant in determining the financial distress of municipalities.

The meaning of these two different components—TCE and EGE—of the overall inadequacy in financial resources can be better understood by taking three different

Italian municipalities as an example (Table 2): Giuliano in Campania, Perugia and Salerno. These three municipalities have almost the same size in terms of population but are affected very differently by the two effects, TCE and EGE, defined above. In particular, in perspective the new equalization transfer mechanism recognizes Giuliano in Campania 89 euros per capita more than the level of resources actually available in 2013 (as mentioned above, the baseline year for the implementation of the equalization) meaning that Giuliano in Campania suffers from a strong inadequacy of resources compared to its SENs. Moreover, the financial resources of Giuliano in Campania were reduced by 23 euros per capita due to vertical transfers cuts carried out in 2014-2015. The case of Perugia is quite different. The new equalization transfers mechanism does not show any resources gap compared to the full financing of its SENs (only 1 euro per capita). However, even Perugia suffered from a cut in vertical transfers by 32 euros per capita. Finally, the case of Salerno is even more favorable. The new equalization mechanism evaluates that the resources available in the baseline year largely exceed what is justified by its SENs, by 172 euros. However, Salerno was also hit by vertical transfers cuts by 50 euros per capita in 2014 and 2015.

As pointed out in Sect. 2, poor management by local policy-makers and officials can be a relevant determinant on the financial distress at municipal level. As a part of the process of estimation of SESs, an extensive data collection was performed via questionnaire sent to all Italian municipalities. The questionnaire asked about expenditure and service levels for several categories of services provided at municipal level. In addition to the evaluation of SENs, the data collected through the questionnaires were used to build a simple system of performance indicators, providing basic information on how each municipality uses available resources for the provision of essential services.<sup>6</sup>

In particular, performance is measured using two specifically calculated indicators. The first is computed for each municipal service as the percentage difference between actual expenditure and the estimated SEN. The elementary indicators for each category of service are then combined into a summary indicator derived as a weighted average across services with weights given by the share of the corresponding expenditure to total expenditure and dividing by municipal population. In what follows we refer to this as the Expenditure Gap (EG). Therefore, a positive (negative) value of EG indicates that the municipality spends more (less) than its standard.

The second performance indicator is computed as the percentage difference between the actual level of services actually delivered (which combines quantity and quality dimensions of the provision) and the estimated standard level of provision consistent with SENs. For example, for waste disposal the quality indicator is the percentage of waste recycled. Also in this case, a summary indicator for all municipal services is derived as a weighted average across services with weights given by the share of the corresponding expenditure to total expenditure and then dividing by municipal population. In what follows we call this as the Output Gap

<sup>&</sup>lt;sup>6</sup> The performance indicators for each municipality are available at: https://www.opencivitas.it/. For more details about the formulation of performance indicators, see Porcelli et al., 2016.

(OG). Therefore, a positive (negative) value of OG points out the ability of the municipality to produce more (less) services than its standard.

However, to provide an adequate measure of the performance at municipal level these two indicators—EG and OG—have to be considered together. Otherwise, for example, a municipality showing a positive EG, that is an expenditure below its standard, could be considered efficient but this is incorrect if it also shows a negative OG, that is a service provision below its standard. Accordingly, for each municipality we defined an overall index of municipal performance simply as the average between OG and EG, following the same approach adopted by the Italian government. This summary index is referred to as Efficiency Index (EI). The higher the value of EI, the higher the municipal performance, that is the ability to produce more services with less expenditure This summary index is included in our empirical model to evaluate the impact of performance of local policy makers and officials on the municipality's probability of incurring financial distress.

For the same sample of municipalities considered as an example to show different combinations of TCE and EGE, Table 3 reports the corresponding values of the EG, OG and EI indices. In particular, even if both Giuliano in Campania and Perugia have a per capita expenditure below the respective standard, only the latter provides an overall level of services higher than the estimated standard and, as a consequence, a positive overall performance (EI>0), whereas in the former low expenditure is associated with poor services for the citizens. Finally, in Salerno the overall performance is undoubtedly negative because, even if spending exceeds its standard, the provision of services is decidedly below its standard.

#### 5 Data and methods

To implement the empirical analysis, we collected an original cross-sectional dataset on all Italian municipalities (6,565 excluding those in Special-Statute Regions), combining information on financial distress indicators in 2016–2019 with information on time-invariant determinants measured at or before 2015. In addition to AIDA-PA, we resorted to different sources of municipal data: the Ministry of Home Affairs dataset as for equalizing transfers under full implementation of the Municipal Solidarity Fund (simulated at the end of the transitional period in 2030) and about transfer cuts implemented in 2014–2015; the Ministry of Economy and Finance (Opencivitas web page) as for the performance indicators measured concerning 2015; the Ministry of Interior and the National Institute of Statistics (ISTAT) as for information on fiscal, electoral, census and morphological characteristics of each municipality as referred to 2015.

Table 8 in the "Appendix" provides a general overview of the descriptive statistics, grouping the variables into three sets: Outcome variables, Main independent variables, and Control variables. As reported in Sect. 3, among Outcome variables the most important indicator is the Financial Rating Index that assigns to each municipality an alphanumeric score over eleven categories from A to E (A, B1, B2, C1, C2, C3, C4, C5, D1, D2, E) that we converted in a numerical score from zero (E) to ten (A). On average between 2016 and 2019 municipalities achieved a score

	Equalization gap effect (EGE)	Transfer cuts effect (TCE)	Expenditure gap (EG)	Output gap (OG)	Efficiency index (EI)
Equalization gap effect (EGE)	1				
Transfer cuts effect (TCE)	0.3139***	1			
Expenditure gap (EG)	-0.1427***	0.0247*	1		
Output gap (OG)	-0.134***	-0.006	0.0499***	1	
Efficiency index (EI)	-0.0047	-0.0215	-0.6465***	0.7298***	1

Table 4 Correlation matrix among main explanatory variables

p values in brackets \*p < 0.10; \*\*p < 0.05; \*\*\*p < 0.001

of 4.8. Other Outcome variables are the already defined Structural Current Equilibrium index, Use of Cash Advances index, and a dummy variable that indicates if the municipality has been subject to legal financial default or intermediate pre-default procedure after 2015. The Main independent variables include the indicators we use as main determinants of financial distress regarding financial resources inadequacy and municipal performance. Among the Control variables, we list fiscal variables such as fiscal effort and expenditure rigidity, political variables that measure the political orientation of the council and the margin of victory of the incumbent mayor in the municipal election, the population dimension and structure by age brackets, morphological characteristics such as the mountain degree and seismic risk. Finally, we also identify if the municipality experienced a council dissolution due to mafia infiltration between 1991 and 2023, information that we used as a proxy for the local level of criminal activity.<sup>7</sup>

As mentioned in Sect. 4, inadequacy of financial resources is measured through the EGE and TCE in the percentage of historical resources. EGE indicates how much the historical resources will change after implementing the equalization period 2030. Therefore, the indicator ranges from negative to positive values; negative values indicate a surplus of resources, and positive values indicate a lack of resources. TCE indicates the amount of grant cuts each municipality has suffered since 2015 (about 10% on average). Municipal performance is measured through OG, EG and EI indices ad in Sect. 4. As mentioned above, the larger the EI, the higher the municipal efficiency, i.e., its ability to produce more services by using less resources.

Table 4 reports the correlation matrix among the main explanatory variables. As expected, the correlation between EGE and TCE is positive and statistically significant. However, the magnitude is 0.31, much below the critical value of 0.7, usually

<sup>&</sup>lt;sup>7</sup> We have decided to keep the entire the whole record of this event since it indicates a higher probability of having organized crime activities going on at the local level independently on the specific year (see, among the others, Ministero dell'interno, 2017).



Fig. 2 Timing and identification strategy

identified as a signal of multicollinearity. Similarly, EGE is also (negatively) correlated, as expected, with EG and OG; however, the magnitude of the correlation (0.14) is very low. Finally, EI, obtained through the combination of EG and OG, is not correlated with EGE (-0.0047 with a *p* value of 0.72) and TCE (-0.0215 with a *p* value of 0.62).

Figure 2 gives a preliminary picture of the distributions of the municipal financial distress indexes across Italian municipalities. What is evident is a strong concentration of high values of the Financial Rating Index and the Structural Current Equilibrium Index in Northern Italy (remember that high values of these indexes correspond to low financial distress conditions). In contrast, the same occurs in Southern Italy when the Use of Cash Advance Index is considered (remember that high values of this index correspond to high financial vulnerability).

To get some initial evidence about the relationship among variables, Fig. 3 reports the correlation plots between fiscal distress indicators and the main determinants we focus on. As for the two measures of inadequacy of financial resources (EGE plus TCE), it is apparent the negative slope of correlation lines (here, the opposite of Use of Cash Advances is considered). So, when EGE and TCE are high, the quality of the municipal budget is low.

The same happens when the two measures of municipal performance are considered (EG and OG). Given the definitions of these indexes, high values of EG are associated with high inefficiency in managing the municipal budget, whereas high values of OG are associated with high efficiency. In brief, the lower the performance



Fig. 3 Financial distress indicators (average 2016–2019)

indexes, the greater the vulnerability of the municipal budget and the more severe the risk of financial distress.

Based on this dataset, the strategy we followed, as illustrated in Fig. 4, is to analyze the effects of the inadequacy in financial resources (EGE and TCE) and of mismanagement (EG, OG and EI), all measured at or before 2015. We aim to provide insights into the financial distress that occurred in Italian municipalities in the immediately following period (average 2016–1019), primarily measured by the Financial Rating Index. For robustness, we also consider the Structural Current Equilibrium Index, the Use of Cash Advances Index, and the legal financial default procedures according to Italian Law.

#### 6 The empirical strategy

The empirical model specified for the estimation of the impact of the lack of resources on financial distress, conditional on the level of efficiency in the provision of services and other controls, is reported in Eq. (1):

$$Y_i = \beta_0 + \beta_1 EGE_i + \beta_2 TCE_i + \beta_3 EI_i + \beta'_4 Controls_i + \epsilon_i$$
(1)

where  $Y_i$  indicates different measures of financial distress (Financial Rating Index, Structural Current Equilibrium, Use of Cash Advances, legal financial default or intermediate pre-default);  $EGE_i$  indicates the EGE (% of 2013 financial resources);  $TCE_i$  indicates the TCE (% of 2013 financial resources);  $EI_i$ : indicates the Efficiency index (% referend to 2015) obtained from the weighted average of  $EG_i$  and  $OG_i$ ; **Controls**<sub>i</sub> is the vector of controls including electoral, census, morphological variables and regional dummies;  $\epsilon_i$  : error component clustered in all specifications at the provincial level.



Fig. 4 Correlations between fiscal distress indicators and main determinants

Given the structure of the dependent variables (some are continuous, and some are discrete), we estimated model (1) by different estimators. In the baseline estimation procedure, we used OLS for all variables except for the dummy that identifies the default or intermediate pre-default; in this case, we estimated a probit model (baseline results are reported in Table 5). Moreover, given the discontinuous structure of the Financial Rating Index, which takes discrete values from 0 to 10, we also estimated a nonlinear model for robustness (results are reported in Table 6). We reclassified the Financial Rating into five categories<sup>8</sup> and estimated an Ordered logit model typically used when the actual values taken on by the dependent variable are irrelevant, except that larger values are assumed to correspond to "higher" outcomes as in our case.

We implemented a heterogeneity analysis to investigate how the main determinants triggering financial distress interact with different municipal characteristics,

<sup>&</sup>lt;sup>8</sup> Score A, B1, and B2 have been classified in the best Category 5; scores C1, C2, and C3 in Category 4; scores C4 and C5 in Category 3; scores D1 and D2 in Category 2, and the worst score E in Category 1.

Jization Gap Effect (EGE), of historical resources ster Cuts Effect (TCE), % historical resources	Rating (0–10) a 2016–2019 –0.0312*** [0.000] 0.103*** [0.000]	werage - 0.00976*** [0.000] - 0.0523***	Indicator R4 Struc equilibrium, avera – 0.000565***	stural current 1ge 2016–2019				
lization Gap Effect (EGE), of historical resources sfer Cuts Effect (TCE), % inistorical resources	-0.0312*** [0.000] 0.103*** [0.000]	-0.00976*** [0.000] -0.0523***	-0.000565***	)	Indicator K8 Us (-1), average 21	e of cash advances 016–2019	Financial defau intermediate pr 2014–2023	lt or e-default
of historical resources sfer Cuts Effect (TCE), % historical resources	[0.000] 0.103*** [0.000]	[0.000] - 0.0523***		-0.000281***	$-0.00188^{***}$	-0.000840***	0.00994***	0.00404***
sfer Cuts Effect (TCE), % historical resources	0.103*** [0.000]	-0.0523 ***	[0.000]	[0.000]	[0.000]	[0:000]	[0.000]	[0.009]
historical resources	[0.000]		0.000751	$-0.00191^{**}$	0.000178	$-0.00704^{***}$	0.0065	$0.0516^{**}$
2 440		[0.003]	[0.294]	[0.011]	[0.914]	[0:000]	[0.670]	[0.027]
iency index (EI), %	$0.266^{***}$	$0.0746^{***}$	0.00375***	$0.00128^{**}$	$0.0139^{***}$	$0.00462^{***}$	$-0.0592^{***}$	$-0.0332^{**}$
	[0.000]	[0000]	[0:000]	[0.024]	[0000]	[0:000]	[0.007]	[0.012]
nator	OLS	SIO	OLS	OLS	OLS	OLS	Probit	Probit
onal dummies	No	Yes	No	Yes	No	Yes	No	Yes
rols (based on 2015 year)	No	Yes	No	Yes	No	Yes	No	Yes
rvations	5994	5994	5994	5994	5994	5994	5994	5994
uared	0.127	0.453	0.039	0.158	0.049	0.156	n.a	n.a
rols (based on 2015 year) ervations uared t estimates with robust stand nditure rigidity, Electoral C anditure Caisanio risck Monure	No 5994 0.127 ard errors cluste ccle, Mayor's pc	Yes 5994 0.453 ited at the provincia altitical orientation, Desident	No 5994 0.039 al level, p values ir The margin of vic	Yes 5994 0.158 1 brackets * $p < 0.10$ for of the incuml	يز في الم	No 5994 0.049 ** <i>p</i> < 0.05; *** nt mayor, the ir on by and brack	No Yes 5994 5994 5994 3.049 0.156 ** <i>p</i> < 0.05; *** <i>p</i> < 0.001. Control v nt mayor, the indication of municij	No Yes No 5994 5994 5994 5994 3.049 $0.156$ n.a ** $p < 0.05;$ *** $p < 0.001.$ Control variables include the mayor, the indication of municipalities dissolved

	(1)	(2)	(3)	(4)	(5)	(9)
	Rating (0–5) aver	age 2016–2019	Rating (0-5) min	2016–2019	Rating (0–5) max	2016-2019
Equalization Gap Effect (EGE), % of	$-0.0214^{***}$	$-0.00834^{***}$	-0.0207	$-0.00817^{***}$	$-0.0212^{***}$	$-0.00791^{***}$
historical resources	[0:000]	[0:000]	[000]	[0:000]	[0000]	[0.000]
Transfer Cuts Effect (TCE), % of	$0.0641^{*}$	$-0.0570^{**}$	$0.0629^{**}$	-0.0442	$0.0728^{**}$	-0.0452*
historical resources	[0.073]	[0.037]	[0.049]	[0.128]	[0.031]	[0.086]
Efficiency index (EI), %	$0.211^{***}$	$0.0566^{***}$	$0.212^{***}$	$0.0647^{***}$	$0.218^{***}$	$0.0535^{***}$
	[0:000]	[0.000]	[0:000]	[0.000]	[0:000]	[0.000]
Estimator	Order Logit	Order Logit	Order Logit	Order Logit	Order Logit	Order Logit
Regional dumnies	No	Yes	No	Yes	No	Yes
Controls (based on 2015 year)	No	Yes	No	Yes	No	Yes
Observations	5994	5994	5994	5994	5994	5994
R-squared	n.a	n.a	n.a	n.a	n.a	n.a
Point estimates with robust standard er Expenditure rigidity, Electoral Cycle, J	rrors clustered at th Mayor's political o	e provincial level, p valu rientation, The margin o	es in brackets $*p < 0.10$ f victory of the incumb	; $**p < 0.05$ ; $***p < 0.0$	01. Control variables in n of municipalities dis	nclude Fiscal eff solved due to ma

 Table 6
 Point estimates of the municipal financial health, rating 0–5, order logit estimator

splitting the sample according to three different municipal features. First, the sample was split between municipalities in Northern and Southern Italy respectively, then between larger and smaller municipalities (above and below 10,000 inhabitants respectively), and finally according to the efficiency level (above and below the median value). We decided not to explore the heterogeneity coming from other factors, such as the political orientation of the local government, given the difficulties in identifying a clear political orientation in small municipals where the mayors are usually representative of local lists.

## 7 Results

As an initial step, we conducted a detailed analysis of how key variables affect the likelihood of legal financial default or intermediate pre-default procedures. These variables include the inadequacy of financial resources (measured by EGE and TCE) and mismanagement (measured by EI). We also included the Financial Rating Index among the regressors to assess whether it effectively predicts defaults when considered alongside other variables. The preliminary findings are presented in Table 10 of the "Appendix". Here, we evaluate the impact on the default likelihood using the following variables, either individually or collectively: Financial Rating Index, EGE, TCE, and EI. The results, as expected, reveal that the Financial Rating Index significantly reduces the default probability. Specifically, an increase in the rating by one unit lowers the default probability by 1.59%, as indicated by the marginal effects in column six of Table 10. When all four variables are analyzed together, the results remain consistent, suggesting that these factors capture different dimensions of financial stress without overlapping.

Table 5 reports the results of the OLS estimation of Eq. (1) considering the three indexes of financial distress separately as dependent variables and, for each of them, excluding or including controls and regional dummies among the covariates (Financial Rating Index in columns (1) and (2); Structural Current Equilibrium in columns (3) and (4); (opposite of) Use of Cash Advance in columns (5) and (6)). The results for the explanatory variables of interest are quite sharp. The estimated coefficients of the indexes of inadequacy of financial resources, EGE and TCE, are statistically significant and show the expected sign for all three financial distress dimensions (particularly when controls and regional dummies are included in the estimated model, as shown in columns (2), (4) and (6)). This means that when the inadequacy of financial resources becomes more severe, the indexes considered here signal a worsening of financial difficulties. The same goes for the mismanagement variable: when mismanagement is high, meaning a low EI (which corresponds to a high EG and low OG), financial distress indexes are high too.

According to our estimates, a one percent increase in the EGE corresponds to a drop of 0.01 points in the Financial Rating Index. Instead, a one percent increase in the TCE corresponds to a more significant reduction of 0.05 points in the Financial Rating Index. Also, taking into account that the average EGE is 20% of the historical resources (considering only municipalities with positive equalization gap) and



Fig. 5 Average marginal effects with 95% confidence interval, determinants (in rows), and Financial Rating index configurations (in columns)

TCE on average is 10% of the historical resources, we still observe that the impact of transfer cuts can be more than twice larger compared to the effects of the equalization gap. This piece of evidence leads us to the critical and intuitive conclusion that a cut in the historical budget, i.e. resources that municipalities were used to spend (also financing inefficient projects), may produce more severe consequences on the financial stability compared to a lack of future funds, i.e. resources that the municipalities should receive but that never were part of the budget.

For completeness, we have also estimated the same model reported in (1) using as a dependent variable a dummy that takes value one if the municipality was subject to financial default or intermediate pre-default in order to estimate the probability of being subject to a legal financial insolvency procedure. The probit coefficient point estimates reported in columns (7) and (8) of Table 5 broadly confirm previous results.

Given our cross-section data, multicollinearity among the main explanatory variables can be an issue. To verify the robustness of our estimate, we computed the Variance Inflation Factor (VIF) indicator that quantifies how much the variance of an estimated regression coefficient increases if your predictors are correlated. A rule of thumb is that if VIF is greater than 10 (some use a lower threshold like 5), multicollinearity is likely a problem, suggesting that the variable is highly correlated with other variables. The VIF of the main explanatory variables is always below 2. Therefore, we think that multicollinearity is not an issue in our model.

As a robustness check, Table 6 reports the coefficient point estimates of the model in Eq. (1), considering the Financial Rating Index reclassified into over five categories as a dependent variable and using the Order logit estimator to see if the discrete structure of the data affects the results obtained using OLS. In Table 6, we show the coefficient point estimates on the main variables of interest (EGE, TCE, and EI) considering three alternative identifications of the dependent variable over

	(1)	(2)	(3)	(4)	(5)	(9)
	Rating (0–10) average 2016–2019 north	Rating (0–10) average 2016–2019 south	Rating (0–10) average 2016–2019 large	Rating (0–10) average 2016–2019 small	Rating (0–10) average 2016–2019 high efficiency	Rating (0–10) average 2016–2019 low efficiency
Equalization gap effect (EGE), % of historical resources	- 0.00839*** [0.000]	-0.0108*** [0.002]	– 0.00644 [0.250]	-0.0114*** [0.000]	– 0.00913*** [0.000]	- 0.0105*** [0.000]
Transfer cuts effect (TCE), %	-0.0554	-0.101*	-0.0655	-0.0513	-0.0489	-0.0648*
of historical resources	[0.107]	[0.089]	[0.324]	[0.124]	[0.183]	[0.075]
Efficiency index (EI), %	$0.0812^{***}$	0.0344	$0.186^{**}$	$0.0676^{***}$	0.0295	$0.0637^{***}$
	[000.0]	[0.184]	[0.021]	[0:000]	[0.300]	[0000]
Estimator	OLS	OLS	OLS	OLS	OLS	OLS
Regional dummies	Yes	Yes	Yes	Yes	Yes	Yes
controls (based on 2015 year)	Yes	Yes	Yes	Yes	Yes	Yes
Observations	4596	1398	426	5568	2997	2997
R-squared	0.308	0.244	0.532	0.454	n.a	n.a
Point estimates with robust sta Rynenditure risidity Flectoral	ndard errors clustered Dvole Mavor's politi	at the provincial level, cal orientation The m	<i>p</i> values in brackets <sup>3</sup> aroin of victory of the	$p_{1}^{0.104} = 0.10; **p < 0.05; ^{3}$	a ***p <0.001. Control vari e indication of municipali	ables include

the 2016–2019 period: the average, the minimum, and the maximum values. For each hypothesis, we specify the model with and without control variables. In total, Table 6 has six columns and, in general, largely confirms the OLS results reported in Table 5. The probability of improving the Financial Rating Index is lower the higher the EGE and the TCE and is higher the higher the E). To visualize the marginal effects on the probability of improving the Financial Rating Index, we have produced Fig. 5, which reports the marginal effects of the three main variables of interest (EGE, TCE and EI) in connection with the three different specifications of the financial index adopted as the dependent variable. Independently on the configuration of the financial index, the probability of being in category five (the highest in terms of financial health) is always negative when EGE and TCE increase by 1%. In line with our original OLS results, TCE has a larger impact than EGE: a 1% increase in TCE reduces the probability of being in category five by roughly 1%. Instead, a 1% increase in the EGE reduces the probability of being in category five only by 0.1%. As expected, the probability of being in category five is always above zero (between 0.5% and 1%) when the EI increases by 1%.

Regarding the heterogeneous effects, the results are reported in Table 7, where we show the OLS point estimates of our three main variables of interest (EGE, TCE, and EI). In particular, columns 1 and 2 show the geographical dimension splitting the sample between municipalities in Northern and Southern Italy, columns 3 and 4 show the sample split between larger and smaller municipalities (above and below 10,000 inhabitants), and finally, columns 5 and 6 split the sample according to the level of efficiency (above and below the median value). The results of Table 6 (confirmed in Table 13 of the "Appendix", where we specify a non-linear model and report order logit point estimates) show that inadequacy of financial resources matters, especially in Southern and/or smaller municipalities; on the other hand, efficiency is important as a determinant of financial distress in Northern and/or larger municipalities. Finally, as expected, increasing efficiency may reduce financial distress mainly if a municipality shows an efficiency index below the median.

For the control variables, a few point estimates (see Table 11 of the "Appendix" for the OLS estimates and Table 12 for Order logit estimates) are statistically significant apart from regional dummies. It is worth noting that the fiscal effort index and the expenditure rigidity of the municipal budget (computed as the ratio of expenditures for personnel to total current revenues) are both statistically significant and show the expected sign: the lower the tax effort and the higher the budgetary rigidity, the higher the financial distress as measured by (almost) all dimensions considered. Results on municipalities dissolved for mafia infiltration show point estimates always with the correct sign, but in rare cases, we observe statistical significance below a p value of 0.10. Lastly, the dummies referred to the Italian regions where municipalities are located are almost all significant for each index of financial distress. The magnitude of the estimated coefficients (nearly always negative given the exclusion of the municipality of region Piemonte) shows a clear divide between the municipalities of Northern and Central Italy (Lombardia, Veneto, Liguria, Emilia-Romagna, Toscana, Umbria, Marche, Lazio, Abruzzo) and Southern Italy (Molise, Campania, Puglia, Basilicata, Calabria): ceteris paribus, financial vulnerability generally increases when moving from Northern to Southern Italy.

## 8 Conclusions

The key question of this paper is whether the conditions of financial distress recently experienced by Italian municipalities can be partially attributed to the inadequacy of financial resources some municipalities suffer compared to the needs of their populations and territories and the mismanagement of local policy-makers and officials and the.

In our investigation, we leverage the novel equalization transfers mechanism, a recent addition to Italy's municipal level. This mechanism offers an exogenous measure of the resource gap that each municipality might be facing, compared to its needs evaluated on a standardized basis (SENs).

The estimation results show that the Italian municipalities suffering from a level of resources below what is necessary to provide public services at standard levels (as measured by a positive Equalization Gap Effect) are ceteris paribus more prone to financial difficulties. By the same token, large cuts in central government transfers (as measured by the Transfer Cuts Effect) have a statistically significant effect on financial vulnerability at the municipal level.

These results, which are based on a multidimensional definition of financial distress, including budget solvency and short-term solvency, confirm and reinforce the evidence we obtained in the case when financial destabilizations of Italian municipalities are detected solely based on the official cases of insolvency regimes envisaged by Italian legislation (structural deficit, intermediate pre-default and financial default). It should be stressed that the budgetary indexes we used in the paper provide a much more informative measure of the financial distress of municipalities than the legal cases of insolvency actually occurred, which can be registered only in qualitative terms (binary variable).

Our findings suggest that financial distress is not only a consequence of economic pressures or administrative mismanagement but is significantly exacerbated by structural imbalances in the allocation of financial resources. The empirical results highlight the dual pressures of drastic cuts in central government transfers and discrepancies in the distribution of funds via new equalization mechanisms. These factors collectively contribute to a scenario where municipalities endowed with insufficient financial resources relative to their service provision standards are predisposed to experience fiscal difficulties. This is compounded by fiscal consolidation strategies that unevenly affect municipalities, exacerbating the fiscal challenges.

From a policy perspective, this analysis underlines the necessity for legislative and fiscal reforms to rectify the structural deficiencies in financial resource allocation. First, it is critical to revisit and refine the criteria and mechanisms for intergovernmental transfers to ensure they are equitable and reflective of the actual expenditure needs as determined by standardized financial health indicators of municipalities. Enhancing the fiscal autonomy of municipalities to allow more localized decision-making regarding revenue and expenditure could also mitigate reliance on central transfers, fostering more sustainable fiscal management. Furthermore, the study underscores the need for a robust fiscal equalization mechanism that effectively bridges the resource gap between economically diverse municipalities. This approach should adjust the amounts transferred and refine the criteria used to determine these transfers to be more responsive to changing economic conditions and fiscal needs.

Additionally, implementing targeted financial support programs for municipalities most at risk of fiscal distress could prevent financial crises. Such initiatives are particularly crucial for regions with historically lower economic performance and higher public service demands. Strengthening financial accountability and management capacities through training and regulatory oversight is also vital in curtailing risks associated with mismanagement and inefficiency, which are significant contributors to financial distress.

Lastly, it is essential to ensure that national fiscal consolidation efforts are wellcoordinated with local government financial planning. Such coordination can help avert situations where local fiscal health is compromised by broader economic policies.

Several possible extensions of the analysis presented in this paper can be outlined. First, it should be more thoroughly investigate the relationship between the legal cases of insolvency and the indexes of financial distress (or a larger array of them) based on municipal budgets. Second, the effects on the financial conditions of the municipality carried out by the two different components of the overall inadequacy of resources we considered—transfer cuts and the equalization gap should be more adequately explored to understand better the way they enter the budgetary decision-making process at the municipal level. In fact, while the cuts to transfers have actually been implemented, the gap in resources highlighted by the equalization mechanism is only virtual. This can produce different effects on the municipal budget through cost stickiness.

## Appendix

See Tables 8, 9, 10, 11, 12 and 13.

Acknowledgements We thank the participants at the 2023 IIPF Annual Congress, the 2023 Annual Conference of the Italian Economic Association, at the 2023 Annual Conference of the Italian Association of Public Economics and at the 2023 Workshop on the Political Economy of Municipal Fiscal Policy for their valuable and insightful comments. Emanuele Padovani (Spoke 2) and Francesco Porcelli (Spoke

=6565)
Regions (n:
Statute
Ordinary
н.
municipalities
statistics,
Descriptive
œ

Table 8         Descriptive statistics, municipalities in Ordinary State	tute Regions $(n = 6565)$				
Variable	Source	Mean	SD	Min	Max
Outcome variables					
Financial rating index (0-10) average 2016-2019	AIDA-PA	4.8979	2.7073	0	10
Structural current equilibrium, average 2016-2019	AIDA-PA	1.0208	0.0851	-0.0250	2.6800
Use of cash advances, average 2016–2019	AIDA-PA	0.1173	0.2540	0.0000	2.3225
Default or intermediate pre-default after 2015	Ca' Foscari Foundation	0.0388	0.1930	0	1
Main independent variables					
Equalization gap effect (EGE), % of historical resources	Ministry of Finance—OpenCivitas	-4.8332	25.0117	- 90.929	230.156
Transfer cuts effect (TCE), % of historical resources	Ministry of Finance—OpenCivitas	10.3328	2.0766	0.0000	22.9008
Expenditure gap (EG), % of Standard—Historical	Ministry of Finance—OpenCivitas	-3.4041	37.4722	- 86.691	2035.46
Output gap (OG), % of Standard—Historical	Ministry of Finance-OpenCivitas	5.2805	43.1095	- 99.546	427.231
Control variables					
Fiscal effort (2015), euro per capita	Ministry of Interior	200.19	336.58	0.00	10,131.74
Expenditure rigidity, average 2016–2019	Ministry of Interior	0.2284	0.0740	0.0000	0.6925
Population × 1000	ISTAT	7.7806	45.7732	0.0290	2872.8000
Electoral Cycle (2015)	Ministry of Interior	3.2888	1.1697	1	5
Mayor's party is left (2015)	Ministry of Interior	0.0938	0.2916	0	1
Mayor's party is right (2015)	Ministry of Interior	0.0954	0.2938	0	1
Mayor's party is center (2015)	Ministry of Interior	0.0052	0.0721	0	1
Margin of victory of the incumbent (2015)	Ministry of Interior	19.976	17.271	0	98
Incumbent at the second term (2015)	Ministry of Interior	0.3838	0.4864	0	1
Municipality dissolved due to mafia infiltration (91-23)	Ministry of Interior	0.0286	0.1667	0	1
Seismic risk	ISTAT	2.7339	1.2432	1	5
Mountain degree	ISTAT	1.8893	0.9537	1	3
Altimetric zone	ISTAT	2.9202	1.5387	1	5

(continued)
ø
Ð
q

Table 8 (continued)					
Variable	Source	Mean	SD	Min	Max
Population 0–2% of total population (2015)	ISTAT	2.5471	0.7353	0.0000	6.1728
Population 3–14% of total population (2015)	ISTAT	10.5486	2.1937	0.0000	17.8438
Population over 65% of total population (2015)	ISTAT	22.8455	5.8257	5.7096	62.6374

Table 9 Italian municipalities           in financial distress 1989–	Year	Financ	cial default	Inter	mediate pre-default
2023 ( <i>Source</i> : Ca' Foscari Foundation)		#	Percentage on total municipalities	#	Percentage on total municipalities
	1989	133	1.68	0	0.00
	1990	68	0.86	0	0.00
	1991	66	0.84	0	0.00
	1992	36	0.46	0	0.00
	1993	37	0.47	0	0.00
	1994	33	0.42	0	0.00
	1995	21	0.27	0	0.00
	1996	4	0.05	0	0.00
	1997	3	0.04	0	0.00
	1998	4	0.05	0	0.00
	1999	5	0.06	0	0.00
	2001	4	0.05	0	0.00
	2002	2	0.03	0	0.00
	2003	2	0.03	0	0.00
	2004	2	0.03	0	0.00
	2005	4	0.05	0	0.00
	2006	3	0.04	0	0.00
	2007	1	0.01	0	0.00
	2008	6	0.08	0	0.00
	2009	6	0.08	0	0.00
	2010	5	0.06	0	0.00
	2011	13	0.16	0	0.00
	2012	15	0.19	47	0.59
	2013	19	0.24	64	0.81
	2014	22	0.28	48	0.61
	2015	18	0.23	30	0.38
	2016	32	0.41	55	0.70
	2017	28	0.35	48	0.61
	2018	30	0.38	45	0.57
	2019	35	0.44	53	0.67
	2020	27	0.34	29	0.37
	2021	22	0.28	44	0.56
	2022	27	0.34	47	0.59
	2023	3	0.04	2	0.03

	(1)	(2)	(3)	(4)	(5)	(9)
Rating (0–10) average 2008–2015	-0.230***				$-0.227^{***}$	$-0.0159^{***}$
	[0000]				[0000]	[0.0017]
Equalization Gap Effect (EGE), % of historical resources		0.00493 ***			$0.00292^{**}$	$0.0002^{**}$
		[0000]			[0.045]	[0.0001]
Transfer Cuts Effect (TCE), % of historical resources			$0.0751^{***}$		$0.0606^{***}$	0.0042**
			[0.000]		[0.007]	[0.0015]
Efficiency Index (EI), %				$-0.0285^{**}$	-0.0260*	-0.0018*
				[0.030]	[0.063]	[6000.0]
Regional dummies	Yes	Yes	Yes	Yes	Yes	Yes
Controls (based on 2015 year)	Yes	Yes	Yes	Yes	Yes	Yes
Observations	5994	5994	5994	5994	5994	5994
R-squared	n.a	n.a	n.a	n.a	n.a	n.a

infiltration, Seismic risk, Mountain degree, Altimetric zone, Resident population, Structure of the population by age brackets

Table 11 OLS point estimates of the municipal fi	inancial health, contro	l variables		
	(1)	(2)	(3)	(4)
	Rating (0–10) average 2016– 2019	Indicator R4—structural current equilibrium, average 2016–2019	Indicator R8 use of cash advances (– 1), average 2016–2019	Financial default or intermediate pre-default 2014–2023
Fiscal effort (2015)	0.000461***	0.0000231***	0.0000130	-0.0000837
	[0:000]	[0.003]	[0.239]	[0.632]
Expenditure rigidity average 2016-2019	$-1.211^{**}$	$-0.197^{***}$	- 0.0585	-0.569
	[0.036]	[0.000]	[0.230]	[0.268]
Population 2015 ( $\times$ 1000)	-0.00176	$-0.000175^{***}$	-0.000203	$0.0150^{***}$
	[0.247]	[0.000]	[0.281]	[0.001]
Population 2015 sq $(\times 1000)$	0.000000552	6.16e - 08***	0.00000109*	$-0.000079^{***}$
	[0.330]	[0.000]	[0.098]	[0.006]
Electoral cycle (2015)	-0.0402	-0.0028	0.00253	0.101
	[0.716]	[0.616]	[0.885]	[0.487]
Electoral cycle sq (2015)	0.0127	0.00059	-0.0000789	-0.0205
	[0.511]	[0.549]	[0.980]	[0.413]
Mayor's party is left (2015)	0.287 **	-0.00539	-0.00124	0.00391
	[0.028]	[0.168]	[0.932]	[0.974]
Mayor's party is right (2015)	0.127	-0.0016	0.00688	-0.0273
	[0.199]	[0.646]	[0.418]	[0.835]
Mayor's party is center (2015)	-0.242	-0.0188*	0.0136	-0.082
	[0.436]	[060.0]	[0.698]	[0.829]
Margin of victory of the incumbent (2015)	0.00275*	0.000101	0.00000367	0.000454
	[0.077]	[0.119]	[0.998]	[0.809]
Incumbent at the second term (2015)	-0.0293	-0.000761	-0.00399	-0.0597
	[0.614]	[0.715]	[0.602]	[0.391]

Table 11 (continued)				
	(1)	(2)	(3)	(4)
	Rating (0–10) average 2016– 2019	Indicator R4—structural current equilibrium, average 2016–2019	Indicator R8 use of cash advances (-1), average 2016-2019	Financial default or intermediate pre-default 2014–2023
Municipality dissolved due to mafia infiltration	- 0.17	-0.0117	0.0446*	0.211
(1991–2023)	[0.222]	[0.314]	[0.054]	[0.121]
Seismic risk	0.0653	0.00124	0.00985	$0.0854^{**}$
	[0.674]	[0.586]	[0.481]	[0.048]
Mountain degree	0.0415	0.00141	- 0.00676	-0.0922*
	[0.632]	[0.487]	[0.148]	[0.082]
Altimetric zone	0.0225	0.00247	0.00373	0.0285
	[0.790]	[0.153]	[0.453]	[0.411]
Population 0–2% of total population (2015)	$0.0933^{*}$	-0.0016	0.00464	0.0576
	[0.080]	[0.531]	[0.514]	[0.421]
Population 3–14% of total population (2015)	$0.0753^{***}$	0.0000575	-0.000224	0.00489
	[0.004]	[0.962]	[0.937]	[0.887]
Population over 65% of total population (2015)	0.00621	0.000227	-0.00131	0.0189
	[0.641]	[0.609]	[0.317]	[0.108]
Lombardia	-0.0339	-0.0120*	-0.0130	-0.115
	[0.888]	[0.081]	[0.203]	[0.449]
Veneto	0.600*	0.0192**	-0.00534	$-0.741^{**}$
	[0.055]	[0.024]	[0.750]	[0.028]
Liguria	$-0.971^{***}$	-0.0227 * * *	-0.0489	$0.544^{***}$
	[0.010]	[0.001]	[0.154]	[0.004]

Table 11 (continued)				
	(1) Rating (0–10) average 2016– 2019	(2) Indicator R4—structural current equilibrium, average 2016–2019	<ul> <li>(3)</li> <li>Indicator R8 use of cash advances (-1), average</li> <li>2016-2019</li> </ul>	<ul><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><li>(4)</li><l< th=""></l<></ul>
Emilia-Romagna	- 0.508 [0.139]	-0.0284*** [0.001]	- 0.0238 [0.219]	- 0.11 [0.638]
Toscana	$-1.771^{***}$	- 0.0184** [0.050]	- 0.0529 10.1001	– 0.125 [0.593]
Umbria	- 3.338***	-0.0339***	- 0.187*** - 0.187	0.302
Marche	[u.uuu] 1.337***	[0.000] - 0.0139	[0.000] — 0.0580**	-0.0453
	[0.001] 2.267***	[0.171] 0.0300***	[0.026]	[0.857] 0.656***
Lazio	- 3.30/*** [0.000]	- 0.000]	[000.0]	[0.000]
Abruzzo	- 2.960*** [0.000]	- 0.0420*** [0.000]	- 0.187*** [0.000]	0.314 [0.118]
Molise	- 2.786*** [0.000]	-0.0324*** [0.000]	-0.232*** [0.000]	0.803*** [0.000]
Campania	- 4.005*** [0.000]	-0.0528*** [0.000]	-0.221*** [0.000]	0.701*** [0.000]
Puglia	- 2.353*** [0.000]	-0.0112 [0.242]	-0.125*** [0.000]	0.633*** [0.000]
Basilicata	- 1.929*** [0.000]	-0.0134* [0.097]	-0.161*** [0.000]	0.217 [0.449]

Table 11 (continued)				
	(1)	(2)	(3)	(4)
	Rating (0–10) average 2016– 2019	Indicator R4—structural current equilibrium, average 2016–2019	Indicator R8 use of cash advances (- 1), average 2016-2019	Financial default or intermediate pre-default 2014–2023
Calabria	$-4.104^{***}$	$-0.0620^{***}$	- 0.204***	0.655***
	[0000]	[0000]	[0.000]	[0.001]
Constant	-0.642	0.981***	$-0.331^{***}$	-0.73
	[0.646]	[0000]	[0.005]	[0.563]
Observations	5994	5994	5994	5994
R-squared	0.453	0.158	0.156	n.a

 $\underline{\textcircled{O}}$  Springer

-		•	
	(1)	(2)	(3)
	Rating (0–5) average 2016–	Rating (0–5) min 2016–2019	Rating (0–5) max
	2019		2016-2019
Fiscal effort (2015)	0.000332***	0.000226***	0.000398***
	[0.000]	[0.002]	[0.000]
Expenditure rigidity average 2016–2019	-0.956*	-0.604	-1.522***
	[0.058]	[0.207]	[0.003]
Population 2015 (×1000)	-0.00128	-0.0017	-0.00193
	[0.474]	[0.250]	[0.200]
Population 2015 sq (×1000)	0.000000489	0.00000051	0.000000812
	[0.434]	[0.340]	[0.115]
Electoral cycle (2015)	-0.0173	-0.0183	-0.178
	[0.883]	[0.866]	[0.184]
Electoral cycle sq (2015)	0.00656	0.0105	0.0362
	[0.757]	[0.572]	[0.126]
Mayor's party is left (2015)	0.224**	0.329***	0.193
	[0.046]	[0.001]	[0.119]
Mayor's party is right (2015)	0.111	0.135	0.0654
	[0.182]	[0.132]	[0.480]
Mayor's party is center (2015)	-0.238	-0.227	-0.422
	[0.463]	[0.295]	[0.119]
Margin of victory of the incumbent (2015)	0.00239	0.0024	0.00291**
	[0.112]	[0.105]	[0.036]
Incumbent at the second term (2015)	-0.0127	-0.0342	-0.0732
	[0.828]	[0.524]	[0.185]
Municipality dissolved due to mafia infiltration	-0.0872	-0.138	0.0163
(1991–2023)	[0.585]	[0.382]	[0.917]
Seismic risk	0.0712	0.07	0.0781
	[0.571]	[0.527]	[0.445]
Mountain degree	0.027	0.0153	0.0183
	[0.771]	[0.938]	[0.914]
Altimetric zone	0.0251	0.0295	0.0373
	[0.695]	[0.583]	[0.446]
Population 0–2% of total population (2015)	0.0868*	0.0595	0.0677
	[0.060]	[0.256]	[0.157]
Population 3–14% of total population (2015)	0.0716***	0.0546**	0.0647**
	[0.002]	[0.011]	[0.012]
Population over 65% of total population (2015)	0.00976	-0.00419	0.0134
	[0.350]	[0.684]	[0.224]
Lombardia	0.00211	-0.112	0.0276
	[0.992]	[0.574]	[0.879]
Veneto	0.520*	0.355	0.585**
	[0.062]	[0.181]	[0.040]

 Table 12
 Point estimates of the municipal financial health, Control variables, Order Logit estimator

	(1)	(2)	(3)
	Rating (0–5) average 2016– 2019	Rating (0–5) min 2016–2019	Rating (0–5) max 2016–2019
Liguria	-0.659**	-0.720**	-0.692**
	[0.025]	[0.018]	[0.012]
Emilia-Romagna	-0.457	-0.484*	-0.429
6	[0.120]	[0.095]	[0.147]
Toscana	-1.333***	-1.342***	-1.437***
	[0.000]	[0.000]	[0.000]
Umbria	-2.530***	-2.776***	-2.728***
	[0.000]	[0.000]	[0.000]
Marche	-1.020***	-1.239***	-0.872**
	[0.001]	[0.000]	[0.013]
Lazio	-2.888***	-3.009***	-2.776***
	[0.000]	[0.000]	[0.000]
Abruzzo	-2.451***	-2.460***	-2.434***
	[0.000]	[0.000]	[0.000]
Molise	-2.357***	-2.415***	-2.268***
	[0.000]	[0.000]	[0.000]
Campania	-3.473***	-3.470***	-3.422***
	[0.000]	[0.000]	[0.000]
Puglia	-1.854***	-1.895***	-1.850***
	[0.000]	[0.000]	[0.000]
Basilicata	-1.528***	-1.553***	-1.527***
	[0.000]	[0.000]	[0.000]
Calabria	-3.673***	-3.543***	-3.584***
	[0.000]	[0.000]	[0.000]
Constant cut 1	-0.426	1.342	-2.092
	[0.720]	[0.263]	[0.143]
Constant cut 2	2.440**	3.772***	1.504
	[0.038]	[0.001]	[0.284]
Constant cut 3	4.054***	5.343***	3.021**
	[0.001]	[0.000]	[0.030]
Constant cut 4	5.900***	7.300***	4.791***
	[0.000]	[0.000]	[0.001]
Observations	5994	5994	5994
R-squared	n.a	n.a	n.a

#### Table 12 (continued)

	(1)	(2)	(3)	(4)	(5)	(9)
	Rating (0–5) average 2016–2019 north	Rating (0–5) average 2016–2019 south	Rating (0–5) average 2016– 2019 large	Rating (0–5) average 2016–2019 small	Rating (0–5) max 2016–2019 high efficiency	Rating (0–5) max 2016–2019 low efficiency
Equalization gap effect (EGE), % of	-0.00588***	$-0.0116^{***}$	-0.00510	- 0.00988***	-0.00728***	$-0.00974^{***}$
historical resources	[0.001]	[0.003]	[0.415]	[0.000]	[0.000]	[0.000]
Transfer cuts effect (TCE), % of	$-0.0558^{**}$	$-0.205^{**}$	-0.0944	-0.0546*	-0.0596*	$-0.0627^{**}$
historical resources	[0.050]	[0.010]	[0.220]	[0.051]	[0.058]	[0.046]
Efficiency index (EI), %	$0.0586^{***}$	0.0313	0.172**	$0.0516^{***}$	0.0117	$0.0476^{***}$
	[0:000]	[0.374]	[0.045]	[0.000]	[0.638]	[0.001]
Estimator	Order Logit	Order Logit	Order Logit	Order Logit	Order Logit	Order Logit
Regional dummies	Yes	Yes	Yes	Yes	Yes	Yes
Controls (based on 2015 year)	Yes	Yes	Yes	Yes	Yes	Yes
Observations	4596	1398	426	5568	2997	2997
R-squared	n.a	n.a	n.a	n.a	n.a	n.a
Point estimates with robust standard e Expenditure rigidity, Electoral Cycle, infilmation Sciemic risk Mountain de	strors clustered at the I Mayor's political orie	provincial level, p valu intation, The margin c Resident nonulation	les in brackets $*p < 0$ of victory of the inco Structure of the nor	0.10; $**p < 0.05$ ; $***p$ umbent mayor, the ind	<0.001. Control varial ication of municipaliti	bles include Fiscal effor es dissolved due to mafi

7) acknowledge funding from the PNRR GRINS project – Growing Resilient, Inclusive and Sustainable (PE00000018 – CUP C93C22005270001), funded by the European Union – NextGenerationEU. The views and opinions expressed are solely those of the authors and do not necessarily reflect those of the European Union, nor can the European Union be held responsible for them.

Author contributions Alberto Zanardi wrote the main manuscript text. Francesco Porcelli ran the econometric exercises. Emanuele Padovani dealt with the institutional analysis. All authors reviewed the manuscript.

Funding Open access funding provided by Alma Mater Studiorum - Università di Bologna within the CRUI-CARE Agreement.

#### Declarations

**Competing interests** The authors declare no competing interests.

**Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicate otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/ licenses/by/4.0/.

## References

- Ahrend, R., Curto-Grau, M., & Vammalle, C. (2013). Passing the buck? Central and subnational governments in times of fiscal stress. OECD Regional Development Working Papers, 2013/05.
- Ahrens, T., & Ferry, L. (2020). Financial resilience of English local government in the aftermath of Covid-19. Journal of Public Budgeting, Accounting & Financial Management, 32(5), 813–823.
- Allers, M. A., & de Natris, J. G. (2021). Preventing local government defaults: No-bailout policy and its alternatives. In R. Geissler, G. Hammerschmid, & C. Raffer (Eds.), *Local public finance: An international comparative regulatory perspective* (pp. 187–208). Springer.
- Ambrosanio, M. F., Balduzzi, P., & Bordignon, M. (2016). Economic crisis and fiscal federalism in Italy. In E. Ahmad, M. Ambrosiano, & G. Brosio (Eds.), *Multi-level finance and the Euro crisis* (pp. 212– 246). Edward Elgar.
- Arachi, G., Porcelli, F., & Zanardi, A. (2023). Redistribution and risk-sharing effects of intergovernmental transfers: An empirical analysis based on Italian municipal data. *Regional Studies*, 57(8), 1523– 1534. https://doi.org/10.1080/00343404.2022.2149726
- Ashworth, J., Geys, B., & Heyndels, B. (2005). Government weakness and local public debt development in Flemish municipalities. *International Tax and Public Finance*, 12(4), 395–422. https:// doi.org/10.1007/s10797-005-2317-3
- Banca d'Italia. (2022). Criticità e prospettive della finanza comunale nel Mezzogiorno alla vigilia del Covid-19. Questioni di Economia e Finanza (Occasional Papers), 708. Retrieved April 27, 2023, from https://www.bancaditalia.it/pubblicazioni/qef/2022-0708/index.html?dotcache=refresh
- Beckett-Camarata, J., & Grizzle, C. (2014). The financial crisis in Harrisburg. *Pennsylvania. Public Finance and Management*, 14(1), 5–29.
- Benito, B., & Bastida, F. (2004). The determinants of the municipal debt policy in Spain. Journal of Public Budgeting, Accounting & Financial Management, 16(4), 492–525. https://doi.org/10.1108/ JPBAFM-16-04-2004-B002
- Bisogno, M., Cuadrado Ballesteros, B., Santis, S., & Citro, F. (2019). Budgetary solvency of Italian local governments: An assessment. *International Journal of Public Sector Management*, 32(2), 122–141.

- Brusca, I., Manes Rossi, F., & Aversano, N. (2015). Drivers for the financial condition of local government: A comparative study between Italy and Spain. Lex Localis—Journal of Local Self-Government, 13(2), 161–184.
- Cabaleiro, R., Buch, E., & Vaamonde, A. (2012). Developing a method to assess municipal financial health. *American Review of Public Administration*, 43(6), 729–751. https://doi.org/10.1177/02750 74012451523
- Cabaleiro, R., & Buch, E. (2011). Impact of size and geographic location on the financial condition of Spanish municipalities. Transylvanian Review of Administrative Sciences, 34E/2011, 22–39. Retrieved May 30, 2024, from https://rtsa.ro/tras/index.php/tras/article/view/297/290
- Capalbo, E., & Grossi, G. (2014). Assessing the influence of socioeconomic drivers on Italian municipal financial destabilization. *Public Money & Management*, 34(2), 107–114. https://doi.org/10.1080/ 09540962.2014.887518
- Carmeli, A. (2008). The fiscal distress of local governments in Israel. Administration & Society, 39(8), 984–1007. https://doi.org/10.1177/0095399707309358
- Carmeli, A., & Cohen, A. (2001). The financial crisis of the local authorities in Israel: A resource-based analysis. Public Administration, 79(4), 893–913. https://doi.org/10.1111/1467-9299.00285
- Chapman, J. (1988). Land use planning and the local budget: A model of their interrelationships. Public Administration Review, 48(4), 800–806. https://doi.org/10.2307/975604
- Cohen, S. (2008). Identifying the moderator factors of financial performance in Greek municipalities. *Financial Accountability & Management*, 24(3), 265–294. https://doi.org/10.1111/j.1468-0408. 2008.00453.x
- Cohen, S., Karatzimas, S., & Naoum, V.-C. (2017). The sticky cost phenomenon at the local government level: Empirical evidence from Greece. *Journal of Applied Accounting Research*, 18(4), 445–463.
- Corte dei conti. (2021). Prime analisi sulla qualità della spesa dei comuni: servizi di amministrazione generale, polizia locale e rifiuti (Deliberazione n. 14/SEZAUT/2021/FRG). Retrieved April 26, 2023, from https://www.corteconti.it/HOME/Documenti/DettaglioDocumenti?Id=b0cc4a45-7439-4e99-a0d7-8c6847f64340
- Corte dei conti (2022). Relazione sulla gestione finanziaria degli enti locali (Deliberazione n. 11/ SEZAUT/2022/FRG). Retrieved April 25, 2023, from https://www.corteconti.it/Download?id= 3a02f00a-884b-4878-be23-8eaece196373
- Cuadrado-Ballesteros, B., Garcia-Sanchez, I. M., & Prado-Lorenzo, J. M. (2013). Determinants of functional decentralization and their relation to debt: Empirical evidence based on the analysis of Spanish municipalities. *International Review of Administrative Sciences*, 79(4), 701–723. https://doi.org/ 10.1177/0020852313501246
- De La Fuente. (2022). La evolución de la financiación de las comunidades autónomas de régimen común, 2002–2020. Estudios sobre la Economía Española, 2022/22. Retrieved April 27, 2023, from https:// documentos.fedea.net/pubs/eee/2022/eee2022-22.pdf?utm\_source=wordpress&utm\_medium= documentos-areas&utm\_campaign=estudio
- Díaz Mendoza, M., Rubio-Ramírez, J. F., González, C. M., & Conde-Ruiz, J. K. (2015). Descomposición de los saldos fiscales en las CC.AA. 2007–2014. Estudios sobre la Economía Española, 2015/2. Retrieved April 27, 2023, from http://documentos.fedea.net/pubs/eee/eee2015-22.pdf
- Garcia-Sanchez, I. M., Mordan, N., & Prado-Lorenzo, J. M. (2012). Effect of the political system on local financial condition: Empirical evidence for Spain's largest municipalities. *Public Budgeting & Finance*, 32(2), 40–68. https://doi.org/10.1111/j.1540-5850.2011.00986.x
- Garcia-Sanchez, I. M., Mordan, N., & Cuadrado-Ballesteros, B. (2014). Do electoral cycles affect local financial health? *Policy Studies*, 35(6), 533–556. https://doi.org/10.1080/01442872.2014.971727
- Gardini, S., & Grossi, G. (2018). What is known and what should be known about factors affecting financial sustainability in the public sector: A literature review. In M. P. Rodriguez Bolivar & M. D. Lopez Subires (Eds.), *Financial sustainability and intergenerational equity in local governments* (pp. 179–205). IGI Global.
- Gorina, E., Maher, C., & Joffe, M. (2017). Local fiscal stress: Measurement and prediction. Public Management Review Spring, 72–94.
- Gregori, W. D., & Marattin, L. (2019). Determinants of fiscal distress in Italian municipalities. *Empirical Economics*, 56, 1269–1281.
- Groves, S. M., Godsey, V. M., & Nollenberger, K. (2003). Evaluating financial condition: A handbook for local government. ICMA.
- Hendrick, R. (2011). Managing the fiscal metropolis: The financial policies, practices, and health of suburban municipalities. Georgetown University Press.

- Jacob, B., & Hendrick, R. (2012). Assessing the financial condition of local governments. In H. Levine, J.
   B. Justice, & E. A. Scorsone (Eds.), *Handbook of local government fiscal health* (pp. 11–40). Jones & Bartlett Learning.
- Jones, S., & Walker, G. (2007). Explanations of local government distress. Abacus, 63(3), 396–418. https://doi.org/10.1111/j.1467-6281.2007.00238.x
- Kimhi, O. (2008). Reviving cities: Legal remedies to municipal financial crises. Boston University Law Review, 88, 633–684.
- Kloha, P., Weissert, C., & Kleine, R. (2005). Someone to watch over me: State monitoring of local fiscal conditions. *American Review of Public Administration*, 35(2), 236–255. https://doi.org/10.1177/ 0275074005277435
- Inman, R. P. (1995). How to have a fiscal crisis: Lessons from Philadelphia. *American Economic Review*, 85, 378–383.
- Levine, H., Justice, B. J., & Scorsone, E. A. (Eds.). (2013). Handbook of local government fiscal health (pp. 11–40). Jones & Bartlett Publishers.
- Lobo, F. C., Ramos, P., & Lourenco, O. (2011). Causes of financial distress of Portuguese municipalities: Empirical evidence. *International Journal of Monetary Economics and Finance*, 4(4), 390–409.
- Maher, C. S., Park, S., McDonald, B. D., III., & Deller, S. C. (2023). Understanding municipal fiscal health: A model for local governments in the USA. Routledge.
- Martell, C. R. (2008). Fiscal institutions of Brazilian municipal borrowing. Public Administration and Development, 28, 30–41. https://doi.org/10.1002/pad.474
- McDonald, B. D. I. I. (2015). Does the charter form improve the fiscal health of counties? Public Administration Review, 75(4), 609–618. https://doi.org/10.1111/puar.12389
- Mikesell, J. L. (2013). Fiscal administration (8th ed.). Wadsworth.
- Ministero dell'interno. (2017). Relazione del Ministro del'interno, Dipartimento per gli affari interni e territoriali sull'attività delle Commissioni per la gestione straordinaria degli enti sciolti per infiltrazione e condizionamenti di tipo mafioso. Anni 2015–2016.
- Navarro-Galera, A., Lara-Rubio, J., Buenida-Carrillo, D., & Rayo-Canton, S. (2017). What can increase the default risk in local governments? *International Review of Administrative Sciences*, 83(2), 397– 419. https://doi.org/10.1177/0020852315586308
- Nemec, J., & Špaček, D. (2020). The Covid-19 pandemic and local government finance: Czechia and Slovakia. Journal of Public Budgeting, Accounting & Financial Management, 32(5), 837–846.
- Padovani, E., Iacuzzi, S., Jorge, S., & Pimentel, L. (2021). Municipal financial vulnerability in pandemic crises: A framework for analysis. *Journal of Public Budgeting, Accounting & Financial Management, 33*(4), 387–408. https://doi.org/10.1108/JPBAFM-07-2020-0129
- Padovani, E., Rescigno, L., & Ceccatelli, J. (2018). Municipal bond debt and sustainability in a nonmature financial market: The case of Italy. *Sustainability*, 10, 3249. https://doi.org/10.3390/su100 93249
- Padovani, E., Scorsone, E., Iacuzzi, S., & Valle de Souza, S. (Eds.). (2022). Local governments' financial vulnerability: analysing the impact of the Covid-19 pandemic (1st ed.). Routledge. https://doi.org/ 10.4324/9781003274278
- Pagano, M., & Moore, R. J. T. (1985). Cities and fiscal choices: A new model of urban public investment. Duke University Press.
- Pammer, W. J. (1990). Managing fiscal strain in major American cities: Understanding retrenchment in the public sector. Greenwood Press.
- Person, C. (2021). Municipalities and excessive debt: Local insolvency regimes as an alternative to bailouts? In R. Geissler, G. Hammerschmid, & C. Raffer (Eds.), *Local public finance: An international comparative regulatory perspective* (pp. 209–227). Springer.
- Porcelli, F., Vidoli, F., Dispotico, R., & Ballanti, D. (2016). Measuring public sector performance: A four quadrants model to monitor local governments' efficiency in Public Sector Economics and the Need for Reforms. MIT Press.
- Raffer, C., & Padovani, E. (2019). Italy. In R. Geissler, G. Hammershmid, & C. Raffer (Eds.), Local public finance in Europe. Country reports (pp. 141–153). Hertie School & Bertelsmann Stiftung.
- Raffer, C., & Ponce, A. (2021). European patterns of local government fiscal regulation. In R. Geissler, G. Hammerschmid, & C. Raffer (Eds.), *Local public finance* (pp. 73–90). Springer.
- Rodriguez-Bolivar, M., Navarro-Galera, A., Alcaide-Munoz, L., & Lopez-Subires, D. (2014). Factors influencing local government financial sustainability: An empirical study. *Lex Localis—Journal of Local Self-Government*, 12(1), 31–54.

- Rodriguez-Bolivar, M. P., Navarro-Galera, A., Alcaide-Munoz, L., & Lopez- Subires, M. D. (2016). Risk factors and drivers of financial sustainability in local government: An empirical study. *Local Gov*ernment Studies, 42(1), 29–51. https://doi.org/10.1080/03003930.2015.1061506
- Solé-Ollé, A. (2006). The effects of party competition on budget outcomes: Empirical evidence from local government in Spain. *Public Choice*, 126(1–2), 145–176. https://doi.org/10.1007/ s11127-006-2456-9
- Tang, S.-Y., Callahan, R. F., & Pisano, M. (2014). Using common-pool resources principles to design local government fiscal sustainability. *Public Administration Review*, 74(6), 791–803. https://doi. org/10.1111/puar.12273
- Ter-Minassian, T. (2007). Fiscal rules for sub-national governments: Can they promote fiscal discipline? OECD Journal on Budgeting, 6(3), 1–11.
- Trussel, J. M., & Patrick, P. A. (2009). A predictive model of fiscal distress in local governments. *Journal of Public Budgeting, Accounting and Financial Management*, 21(4), 578–616. https://doi.org/10. 1108/JPBAFM-21-04-2009-B004
- Wallstedt, N., Grossi, G., & Almqvist, R. (2014). Organizational solutions for financial sustainability: A comparative case study from the Swedish municipalities. *Journal of Public Budgeting, Accounting* & Financial Management, 26(1), 181–218.
- Wang, X., Dennis, L., & Tu, Y. S. (2007). Measuring financial condition: A study of U.S. states. Public Budgeting & Finance, 27(2), 1–21.
- Zafra-Gomez, J. L., Rodriguez-Bolivar, M. P., & Munoz, L. A. (2013). Contrasting new public management (NPM) versus post-NPM through financial performance: A cross-sectional analysis of Spanish local governments. *Administration & Society*, 45(6), 710–747. https://doi.org/10.1177/0095399711 433696

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.